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ORLANDO PARTNERING TEAM - MEETING bWWTES-~-- ---------- 

Date: 2,3 October 2000 
Location: Oak Ridge, TN 
Team Leader: David Grabka 
Recorder: Wayne Hansel 
Gatekeeper/Timekeeper: Steve McCoy 
Facilitator: Stephanie Fraser-Beekman (email: drsbeekman @ aol.com) 

ATTENDEES 

OPT MEMBERS: 
Rick Allen 
David Grabka 
Wayne Hansel 
Steve McCoy 
Nancy Rodriguez 
Steve Tsangaris 

SUPPORT MEMBERS: 
Barbara Nwokike 

GUESTS: 

HANDOUTS DISCUSSED AT THE MEETING 

1. UST/IR Update and Status - October 2000 (Rick) 
2. SA 40 arsenic concentrations in Soil - Steve M. 
3. SA 54 sample results - Steve M. 

2 October 2000 

CHECK-IN 

Everyone was Great and Life is Good! 

Action Items 

Went through action items (see revised list at end of minutes). 

UST/IR/Transfer Updates 

UST/lR: See handout (updates are bold/italicized in the handout) 

Transfer: 
l Transferred about 80 acres to GOAA 
l Working on deeds for EDC conveyance to the City of Orlando at Herndon and McCoy. 
o ESS and EBST for Area C is being reviewed by the VA. 
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OU 2 DPT results 

Steve M. gave short presentation on detection of VOCs in DPT points just 
resampled. There is a Benzene plume across the South end of landfill cells and TCEi plume at 
well 18B. Action Item: Steve to propose where to take NA samples. Decision: Install 
permanent well to confirm Benzene plume levels at push P5 location. 

Tier It Update 

Tier I & Tier II meeting is on (Possible March timeframe). Action Item: Bring problems needing 
Tier II attention. 

OU 2 RI response to comments 
Mike Campbell discussed comments on OU2 RI Report. Written comments will be out soon. 
EPA still has not received comments from Eco review. Action Item: Nancy to get status on 
report comments. 

2 October 2000 

Training 

Stephanie led team accomplishment and evaluation. 

Study Area 39,36,40,54 Update 

SA 39 Veggy Oil Injection. Finalizing details for injection design. May use gravity or diaphragm 
pump (10 - 15 psi) Variance is not needed, just notify UIC. Horay Casperie will be the new 
review engineer for FDEP. David needs more time to review workplans. 

SA 36 Veggy Oil Injection. Downward groundwater gradient in middle of the site. Still do not 
have clean up and down gradient wells. Steve T. will install up and down gradient wells when 
work is done to install injection wells in November. Action Items:.Steve M Propose location of 
up and down gradient wells. 

SA 40 results were handed out. Mixing only reduced levels to residential in about 30% of area. 
Team decided to remix site. Action Item: Steve and Steve to coordinate and schedule remix. 

SA 54 sample results were handed out. Still donut have clean line. 
Action Item: Steve M. was tasked to completed delineation and plot PAH equilvants at site. 

OU 4 Interception wells 

Workplans are out, stripper is on order will be in early November. 

IDW 

Action Item: Barbara to add drum testing to Steve M SOW. 
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SA 17 update 

Steve T. Workplans are out. Injection planned 18 - 19 or 23 - 30 October. 

OU 1 Monitoring 

Steve T. discussed Summary of Semi-Annual Monitoring for OUl. Not much change in data. 
Report was issued August 30,200O. 

- 
OU 3 Treatability Study (Column Studies) 

With retardation factor K factor, plume movement may only be IO inches per year. The final 
column test should be completed by our next meeting. Activated Alumna was the definite 
winner. 

SA 2 ORC 

Steve M. POA is in SOUTHDIV ready to award. Fieldwork to start in November. Bench scale is 
the first thing that will be done. 

OU 4 permanganate study, FS costs for OU 4 

Final pilot study report almost completed. Action Item: Rick Allen to check on monitoring 
intervals after shutoff. 

.-Critique, Agenda, Close-Out 

+‘s 
Barbara on time 
Fishing & Dinner 
Weather 
Free Desert 
Training in fresh air 
Good leader David was Upbeat not beatup 

Catfish not biting 
Starter problems 

A’S 

Future Meeting Schedule 

November 15 and 16 - Orlando 
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0711 o/o0 

ACTION ITEM SUMMARY 
October 2000 

Action Items (Carrvover) 

3. 

Steve M., Steve T., and Rick will coordinate to complete schedule. Final will be distributed 
to the OPT for site management, business planning purposes, and developer coordination. 
Final will be uploaded to David PullmanKH2M HILL to compile in the master schedule for 
SouthDiv. Status: Done 

Steve M.: When data are available, revise SA35 SSR, update report and revise a.s per 
comments received during the meeting. Status: Ongoing. Data are in, just need to put 
together report. 

Wayne to talk to property owners at SA 2 about contaminated groundwater. Status: On 
hold 

Nick: Send BAP data from SA 16 ditch to Dave. Status: Ongoing. 

Steve M., Produce lithologic sections at SA 36. Status: Ongoing. 

Action Items (New) 

Steve M., Propose were to take NA samples for OU 2. 

Everyone, Bring problems needing Tier II attention. 

. Nancy, Get status on report comments for OU2 IR report. 

4. Steve M., Propose location of up and down gradient wells at SA 36. 

5. Steve and Steve, Coordinate and schedule remix of SA40. (Chp.. 

Steve M. was tasked to completed delineation and plot PAH equilivants at site for SA54. 

arbara, Add drum sampling to Steve M. SOW 

Rick Allen; Check on monitoring intervals after shutoff. 
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